Feed the Future Innovation Lab for Food Security Policy

Research Paper 150

September 2019

CAN CONSERVATION AGRICULTURE SAVE TROPICAL FORESTS? THE CASE OF MINIMUM TILLAGE IN ZAMBIA

By

Hambulo Ngoma and Arild Angelsen











Food Security Policy Research Papers

This *Research Paper* series is designed to timely disseminate research and policy analytical outputs generated by the USAID funded Feed the Future Innovation Lab for Food Security Policy (FSP) and its Associate Awards. The FSP project is managed by the Food Security Group (FSG) of the Department of Agricultural, Food, and Resource Economics (AFRE) at Michigan State University (MSU), and implemented in partnership with the International Food Policy Research Institute (IFPRI) and the University of Pretoria (UP). Together, the MSU-IFPRI-UP consortium works with governments, researchers and private sector stakeholders in Feed the Future focus countries in Africa and Asia to increase agricultural productivity, improve dietary diversity and build greater resilience to challenges like climate change that affect livelihoods

The papers are aimed at researchers, policy makers, donor agencies, educators, and international development practitioners. Selected papers will be translated into French, Portuguese, or other languages.

Copies of all FSP Research Papers and Policy Briefs are freely downloadable in pdf format from the following Web site: <u>https://www.canr.msu.edu/fsp/publications/</u>

Copies of all FSP papers and briefs are also submitted to the USAID Development Experience Clearing House (DEC) at: <u>http://dec.usaid.gov/</u>

AUTHORS

Ngoma: Department of Agricultural Food and Resource Economics, Michigan State University, East Lansing, MI 48824-1039, USA, Indaba Agricultural Policy Research Institute (IAPRI) Lusaka, Zambia and School of Economics and Business, Norwegian University of Life Sciences (NMBU), PO Box 5003, 1432 Ås, Norway & Center for International Forestry Research (CIFOR), Bogor, Indonesia.

Angelsen: School of Economics and Business, Norwegian University of Life Sciences (NMBU), PO Box 5003, 1432 Ås, Norway, and & Center for International Forestry Research (CIFOR), Bogor, Indonesia.

ACKNOWLEDGMENTS

We gratefully acknowledge funding from the Norwegian Agency for Development Cooperation [agreement no. GLO-3945 QZA 13/0545; grant on Learning from REDD+ - an enhanced Global Comparative Analysis] through the Center for International Forestry Research (CIFOR). The first author acknowledges financial support provided by USAID through the Zambia buy-in to the Feed the Future Innovation Lab for Food Security Policy (FSP) under Cooperative Agreement No. AID-OAA-L-13-00001. We thank Davison Gumbo for hosting and facilitating fieldwork and all research assistants. We are grateful for comments from the EAERE 2016 conference in Zurich and the first author's Ph.D. thesis evaluators, and to John Herbert Ainembabazi. The contents are the sole responsibility of the authors and do not necessarily reflect the views of NoRAD, USAID, CIFOR or NMBU.

This paper is also published as an Norwegian University of Life Sciences School of Economics and Business Working Paper No. 2/ 2017 and a journal article in *Forest Policy and Economics* Volume 97, December 2018, Pages 153-162.

This study is made possible by the generous support of the American people provided to the Feed the Future Innovation Lab for Food Security Policy (FSP) through the United States Agency for International Development (USAID) under Cooperative Agreement No. AID-OAA-L-13-00001 (Zambia Buy-In). Additional funding support was also provided by the Swedish International Development Agency (SIDA) and the USAID Mission to Zambia. The contents are the sole responsibility of the authors and do not necessarily reflect the views of USAID, the United States Government, SIDA, or MSU.

Copyright © 2019, Michigan State University IAPRI All rights reserved. This material may be reproduced for personal and not-for-profit use without permission from but with acknowledgment to MSU and IAPRI.

Published by the Department of Agricultural, Food, and Resource Economics, Michigan State University, Justin S. Morrill Hall of Agriculture, 446 West Circle Dr., Room 202, East Lansing, Michigan 48824, USA.

ABSTRACT

Minimum tillage (MT) is a key component in the promotion of conservation agriculture (CA). This paper asks whether MT reduces cropland expansion and thus deforestation. We develop a simple theoretical household model of land expansion, and test hypotheses by estimating a double hurdle model using household survey data from 368 smallholders in rural Zambia. We find that about 19% of the farmers expanded cropland into forests, clearing an average of 0.14 ha over one year. Overall, MT adoption does not significantly reduce deforestation among households in our sample, while labor availability stimulate expansion. Yield augmenting agricultural technologies (such as MT) may not reduce expansion unless combined with other forest conservation measures.

Keywords: Cropland expansion, deforestation, minimum tillage, double hurdle, Zambia

ABSTRACT	iv
LIST OF TABLES	vi
LIST OF FIGURES	vi
LIST OF ACRONYMS	vii
1. INTRODUCTION	1
 THEORETICAL MODEL	3 3 5
 EMPIRICAL STRATEGY	7 7 7 9
4. DATA SOURCES	10
 RESULTS	12 12 13 14
 6. DISCUSSION	17 17 17
7. CONCLUSION	19
REFERENCES	20
APPENDICES	24
A. Proof of Proposition	24
B. Endogeneity Tests, Robustness Checks and Production Function Estimates	31

CONTENTS

LIST OF TABLES

TABLE	PAGE
 Variables Used in Regression Models and Summary Statistics	11 opland
into the 2013/2014 Season and Those That Did Not	
3. Double Hurdle Average Partial Effects of Factors Influencing Cropland Expansion	15
B 1. First Stage – Reduced Form OLS Estimates for the Endogenous Variables	31
B 2. Coefficient Estimates from Double Hurdle Models with Control Function-IV Approa	iches32
B 3. Cobb Douglas Production Function Estimates	

LIST OF FIGURES

PAGE

FIGURE

1 Spatial Location of Survey Areas in Zambia	10
2 Reasons Earmers Expanded Cropland into the 2013/2014 Season	12
3 Quadratic Predictions of the Effects Minimum Tillage Adoption Intensity (A) Yield (B) Farm	1 4
Size (C) and Value of Household Assets (D) on Cropland Expansion	14
She, (c) and value of Household Assets (D) on orophand Expansion	

LIST OF ACRONYMS

- APEs Average Partial EffectsCA Conservation Agriculture
- CSA Climate Smart Agriculture
- FAO Food and Agricultural Organization
- FOCs First Order Conditions
- GHG Greenhouse Gases
- ha hectare
- kg kilogram
- km kilometer
- MT Minimum Tillage
- SSA Sub-Saharan Africa
- USD U.S. Dollar
- ZMK Zambia Kwacha (old)
- ZMW Zambia Kwacha

1. INTRODUCTION

Smallholder farmers in Sub-Saharan Africa (SSA) need to raise agricultural productivity to feed a growing population, and – relatedly – to increase farm income to reduce own poverty. Costly agricultural inputs and constrained access to credit leave area expansion as the main option for many smallholders. Despite its potential to improve both crop yields (Ehui and Hertel 1989) and production (De Janvry and Sadoulet 2010) in the short-term, cropland expansion is often unsustainable because farmers face diminishing land productivity and adverse environmental impacts of forest-loss.

Agricultural land expansion is the main cause of tropical deforestation (Angelsen and Kaimowitz 2001; Gibbs et al. 2010). Forest loss has contributed about one-third of the accumulated increase in greenhouse gases (GHG), and makes up about one-tenth of the current emissions (IPCC 2013). Climate change exposes smallholders to higher rainfall variability and other climate related shocks (Hallegatte et al. 2016). This hampers the efforts to reduce poverty in SSA, both due to the direct effects on crop production, and indirectly through destabilizing agricultural markets and higher risks making farmers reluctant to undertake investments in the sector. The dual challenge of smallholder farmers in SSA is, therefore, to intensify agricultural production sustainably while mitigating and adapting to predicted climate change.

Several agricultural technologies have been proposed to meet this dual challenge, based on the assumption that intensifying agriculture would raise land productivity and spare nature. These technological options, exemplified by the Asian green revolution of the 20th century, altered factor intensities by increasing the use of labor, capital, inorganic fertilizers, improved seeds and tillage. Currently, there is strong political support for climate smart agriculture (CSA) as a major avenue to simultaneously raise smallholder agricultural productivity and enhance climate change adaptation and mitigation in SSA. CSA is a broad-based approach that includes policy reforms to support new technological solutions and farm management practices, which includes conservation agriculture (CA). CA has three key components: minimum tillage (MT), *in-situ* crop residue retention and crop rotation (Thierfelder et al. 2015). It aims to improve agricultural land productivity while delivering adaptation-mitigation co-benefits (Govaerts et al. 2009; IPCC 2014; Thierfelder and Wall 2010; UNEP 2013).

This paper focuses on household-led cropland expansion and addresses the where, why and how much questions. We test the effects of CA practices on cropland expansion using a consistent theoretical household model and an empirical analysis based on detailed survey data on farm, household and contextual characteristics. We focus on MT because it is the basis and main component of CA. Our data are from Zambia, a country which – despite a high forest cover of about 66% (FAO 2015; Kalinda et al. 2013) – experiences high deforestation driven by household or centrally planned (e.g., farm blocks) agricultural expansion, urbanization, new settlements, road development and mining.

The literature suggests two major pathways through which MT may contribute to reduced GHG emissions. The first is through improved soil carbon sequestration resulting from reduced tillage and enhanced buildup of soil organic matter (UNEP 2013). However, a growing evidence base suggesting that MT has limited potential to sequester soil carbon challenges this view (Powlson et al. 2014; Powlson et al. 2016; VandenBygaart 2016). The second pathway, the focus of this paper, links MT to reduced deforestation via its effects on crop yield and input use.

MT improves crop yield by facilitating early planting, buildup of soil organic matter and improved input use efficiency (Haggblade and Tembo 2003). On one hand, higher yield means the same

output can be delivered from less agricultural land, and should therefore take the pressure off forests, sometimes referred to as the Borlaug hypothesis (Angelsen and Kaimowitz 2001). However, higher yield also provides an incentive to shift resources to a more profitable agricultural sector, including expanding the land area if feasible. Angelsen and Kaimowitz (2001) found that new technologies that raise agricultural productivity might in fact stimulate deforestation by making agriculture more profitable, sometimes referred to as the 'Jevons' paradox. The constant output assumption does not hold.

MT is also labor-intensive (Giller et al. 2009; Thierfelder et al. 2015), especially in the initial years of adoption. Labor-intensive technologies absorb family labor (and possibly raise rural wages) and might therefore have a land-sparing effect. As such, it is not readily clear how MT affects deforestation. This depends on the factor intensities and yield impact of the particular MT technology, market conditions, and preferences and farm constraints (Angelsen 1999). The scale of the analysis also matters. Large-scale adoption provokes general equilibrium effects in the form of lower output prices and higher wages (if labor intensive), which can partly or fully offset the effect of higher profitability (Angelsen and Kaimowitz 2001).

Most studies on the link between agricultural technologies and deforestation are global or national in scope and apply inadequate econometric methods (Barbier and Burgess 2001; Gibbs et al. 2010; Phelps et al. 2013; Rudel et al. 2009; Rudel 2013), making them less informative to local contexts. Others lack explicit theoretical models to guide their empirical analysis, and many focus only on output effects (Balmford, Green, and Scharlemann 2005; Ewers et al. 2009). Except for Vinya et al. (2011) and Holden (1997), empirical evidence on the nexus between agricultural technologies and deforestation remains thin in Zambia.

This paper adds to the literature in two ways. First, we develop a simple Chayanovian agricultural household model and solve it analytically to guide the empirical estimation. Second, we address data problems by using detailed local context-household survey data from rural Zambia for the empirical analysis and we account for the potential endogeneity of MT, shadow wages and yield on cropland expansion decisions using instrumental variables.

The rest of the paper is organized as follows. Section 2 briefly reviews existing models of deforestation and presents the theoretical model. Sections 3 and 4 outline the empirical strategy and data sources, while section 5 presents the main results. Section 6 discusses these results, and section 7 concludes.

2. THEORETICAL MODEL

2.1 Economic Models of Deforestation

Theoretical economic models of deforestation can be classified in two broad categories. The first one applies dynamic optimization on the allocation of land between forest and competing uses (mainly agriculture), possibly also including different sub-sectors of agriculture, e.g., lowland and upland agriculture. With a national or region focus, these models assume that a social planner determines land allocation by comparing relative returns over time. Examples include Barbier and Burgess (1997) and Tachibana, Nguyen, and Otsuka (2001). In general, these models show that higher agricultural profit stimulate deforestation by favoring conversion of land to agriculture. The idea of a social planner determining land allocation may be far-removed in countries with partly liberalized land markets like Zambia. Yet a market system can mimic some of the characteristics of the optimal solution, but additional features need to be added, e.g., insecure tenure and land claims strengthened by land clearing.

The second category focuses on household level drivers of deforestation and apply different versions of the (mostly) static agricultural household model in the tradition of Singh, Squire, and Strauss (1986). These models are again split between recursive and non-recursive models, i.e., based on whether consumption and production decisions can be separated (recursive) or must be taken simultaneously. The former assume that households participate in well-functioning labor, land and output markets, and production decisions are studied within a profit-maximizing framework. Missing or imperfect markets give rise to the non-recursive models, also labelled Chayanovian models. (See Angelsen (1999) and Angelsen (2010) for a comparison of different models of deforestation, and Pagiola and Holden (2001), Maertens, Zeller, and Birner (2006), and Alix-Garcia, Shapiro, and Sims (2012) for other applications of agricultural household models on deforestation.)

Our theoretical model falls within the second category of deforestation models, i.e., a static model assuming an imperfect labor market (non-recursive model).

2.2 Farm Level Cropland Expansion

We develop a Chayanovian model of cropland expansion for a representative, smallholder farm household in Zambia. We extend existing models of, *inter alia*, Angelsen (1999) and Maertens, Zeller, and Birner (2006) by adding a new technology (MT). We assume that land is homogenous, available and accessible at a cost $d(A - A_0)$. We assume a well-behaved aggregate production function, in which production is a function of family labor (l^a) , minimum tillage (M), land area (A), and inputs (X): $Y = f(l^a, M, A; X)$. We consider M a non-essential input. The vector of inputs X is assumed fixed, which enables us to solve the model analytically without too much complication. Labor, minimum tillage and land are assumed to be complementary: $f_{l^a}, f_A, f_M > 0$; $f_{l^al^a}, f_{AA}, f_{MM} < 0$; $f_{l^aA}, f_{l^aM}, f_{MA} > 0$ and $f_{AM} = f_{MA} = 0$. Given the complementarity assumption, agricultural land expansion and MT adoption have implications for the demand of family labor.

A representative household maximizes utility U = U(c, l; h) by trading off consumption (c) and leisure time (l). h is a vector of exogenous household demographics, which we drop in subsequent equations beyond Eq. (1) but retain in the empirical estimation. In addition to land access cost (d), we assume an additional convex labor cost of clearing, reflecting, for example, increasing costs with distance from the homestead as crop area expands: $t(A - A_0)^r$, t > 0 and r > 1. A is total cultivated land per household and A_0 is the initial stock of land. The difference $(A - A_0)$ then gives the size of converted land within this period. A representative farm household solves the following problem:

$$\underset{c,l^{a},A,M}{Max U} = U(c,l;\boldsymbol{h})$$
(1)

Subject to

$$c = p_{y} f\left(l^{a}, M, A; \boldsymbol{X}\right) - vM - d\left(A - A_{o}\right) - p_{x}\boldsymbol{X} + w\overline{l^{o}} + E$$
⁽²⁾

$$T = l + t \left(A - A_o \right)^r + l^a + \overline{l^o}$$
(3)

$$c, A, l^a > 0; M \ge 0; r > 1; t > 0; A \ge A_a$$

U is household utility, which has positive but diminishing marginal utilities of consumption and leisure. For simplicity, we assume zero cross partials: $U_c, U_l > 0; U_c, U_u < 0; U_c, U_d = 0$. Eq. (2) is the budget constraint. We assume that all income earned by the household is spent on consumption in our single period model. p_y is output price, v is the farm level input and capital costs of implementing MT. p_x is the per unit input cost, which like X, is fixed. E captures all other exogenous income to the household. Eq. (3) is the time or labor constraint; total household time T equals leisure time l plus time spent working on the farm l^a , clearing new agricultural land $t(A - A_o)^r$ and off the farm l^o .

We assume an imperfect labor market in the sense that households can only sell a fixed amount of labor \tilde{l}^{o} at a market wage rate *w*. This labor market assumption leads to the Chayanovian model with a household specific shadow wage below the market wage (otherwise, the constraint will not be binding, see Angelsen (1999)). We assume that output markets are well functioning in the way that farmers can sell (or buy) what they want at a given market price.

We consider the interior solution case where $A - A_0 > 0$ and M > 0, and write the Lagrangian for the problem as:¹

$$L = U(c,l) + \mu \Big[c - p_{y} f(l^{a}, M, A; X) + vM + d(A - A_{o}) + p_{x} X - wl^{o} - E \Big] + \lambda \Big[T - l - t(A - A_{o})^{r} - l^{a} - \overline{l^{o}} \Big]$$
(4)

Solving Eq. 4 and after some manipulations, the first order conditions (FOCs) for the choice variables l, c, M and A are given by:

$$p_{y}f_{l^{a}} = z \tag{5}$$

$$p_{y}f_{A} = d + zrt\left(A - A_{o}\right)^{r-1}$$

$$\tag{6}$$

$$p_{y}f_{M} = v \tag{7}$$

¹ We consider cases where M = 0 in the empirical estimation.

Where

$$z = \frac{U_l(c,l)}{U_c(c,l)}.$$
(8)

Eq. (8) defines the shadow wage (z) as the marginal rate of substitution between consumption and leisure. Eq. (5) states that the marginal productivity of agricultural labor (or leisure) is equal to z. The marginal productivity condition in Eq. (6) states that a household will expand cropland (A) until the marginal productivity of land equals the sum of the cash and labor cost of land and expansion. Eq. (7) states that MT is profitable as long as its marginal benefit is equal to the cost of implementing it.

Since MT (M) and cropland expansion (A) are both endogenous, assessing the impact of MT on expansion cannot readily be seen from the FOCs without further comparative statics. Also, since M is endogenous in the model, the exogenous (policy) variable to investigate in our formulation is the costs of implementing MT, namely v. As seems from Eq. (2), a policy lowering v will lead to higher adoption of MT.

2.3 Comparative Statics

The complete comparative statistics is presented in Appendix A. We used Cramer's rule to assess how changes in exogenous variables affect cropland expansion (A), the key outcome variable of interest. This section only discusses results for the effects of the cost of implementing MT, (v) on cropland expansion (A): $\partial A / \partial v$. Appendix A reports the full derivations.

The impact of changes in (costs of) MT adoption on land expansion is the net effect of the substitution and income effects. The substitution effect is what a recursive model would give, i.e., by keeping z constant. The income effect is analyzed through changes in in z (Angelsen (1999).

The substitution effect of a change in v on A is as follows: lower (higher) costs of MT adoption increases (reduces) adoption. Higher M increases the marginal productivity of land, given the complementarity assumption. Expansion becomes more profitable, and the substitution effect only gives that $\partial A / \partial v$ is negative.

The income effect is, however, positive. Lower v has several effects on z. First, all inputs remaining constant, lower v reduces the costs of production and raise consumption (cf. Eq. (2)). Higher consumption raises z, cf. Eq. (A1). Second, the substitution effect gives higher M, A and l' when v is lowered. The household will both produce and work more, and have less leisure. This triggers a further increase in z, cf. again Eq. (A1). The income effect will therefore have a negative impact on expansion though the higher shadow wage rate, i.e. positive impact on $\partial A / \partial v$.

We summarize our results; cf. Eq. (A31) in the following proposition:

Proposition: The overall effects of MT on cropland expansion are indeterminate, *a priori*. The substitution effect is positive, reflecting the higher profitability of land expansion. The income effect is negative, reflecting that higher consumption and less leisure raise the household shadow wage rate, reducing the profitability of land expansion.

We cannot determine the net effect, but can hypothesize how different factors will affect the net result. Household preferences matters, i.e., responsiveness of the shadow wage to changes in consumption and leisure. Angelsen (1999) shows, using a Stone Geary utility function with

subsistence levels, that the income effect dominates for poor households (close to that subsistence level). The production technology also matters, e.g., to what extent MT adoption changes the marginal productivity of labor and land. If MT adoption leads to large increases in marginal labor productivity (i.e., is labor intensive), the impact on z can be large, and make the income effect dominate the substitution effect.

3. EMPIRICAL STRATEGY

3.1 Reduced Form Equation

In order to bring the theoretical framework to the data, we can write a parsimonious representation of the reduced form solution as: ²

$$\boldsymbol{A}^{*} = \boldsymbol{A}^{*} \left(\boldsymbol{M}, \boldsymbol{y}, \boldsymbol{z}; \boldsymbol{p}_{\boldsymbol{y}}, \boldsymbol{p}_{\boldsymbol{x}}, \boldsymbol{X}, \boldsymbol{A}_{o}, \boldsymbol{h} \right) + \boldsymbol{\varepsilon},$$

$$\tag{9}$$

where \mathcal{A}^* is the size of expanded cropland, M is the size of land under MT. We use the size of land area under MT, which also reflects the cost of implementing MT (v). The direct cost of implementing MT (v) was not collected during the survey. y is aggregate crop yield (used as proxy for expected yield), z is the shadow wage (opportunity cost of labor). p_y is the per kg cost of maize used as a proxy for output price.

X is now a dummy = 1 if a household used inorganic fertilizers and/or hybrid seed, p_x is an average per kg input cost of inorganic fertilizer and seeds, l^o is a dummy = 1 if at least one household member worked off farm, and A_o is the initial amount of land (farm size) controlled by the household (net of newly expanded area). h is a vector of demographics and other exogenous variables: value of assets, subsidies, tenure, distance from homestead to protected forest and location dummies. ε is the error term. Table 1 gives detailed descriptions and summary statistics for all variables in Eq. (9).

Following from the theoretical model, M, y, and z are endogenous and therefore may be correlated with the error term in Eq. (9). There are also reasons to suspect that MT is endogenous because farmers who self-select themselves into MT adoption may have unobservable characteristics, which may also influence their expansion decisions. Moreover, since MT affects cropland expansion through yield effects, then MT and yield may be interdependent, and possibly jointly determined with expansion decisions at household level. Subsection 3.1 elaborates how we addressed these concerns.

We computed the household specific shadow wage (z) following Jacoby (1993). First, we estimated a Cobb Douglas production function and then used the results (Table A3, appendix B) to compute:

$$z = \beta \frac{prod}{labor} \tag{10}$$

where *labor* is total labor input per household, β is an estimated parameter associated with labor and *prod* is the predicted value of production (the dependent variable in the estimated production function).

3.2 Estimation Challenges

A natural first step before deciding how to estimate Eq. (9) is to test for the endogeneity of M, y and z. We used the control function approach of Wooldridge (2010). This involves estimating reduced form equations for each endogenous variable as the dependent variable regressed on all exogenous variables and an additional instrumental variable(s) (IV).

² This representation relaxes some assumptions from the theoretical model for example on fixed p_x and X

The reduced form equations were of the form:

$$endog_{i} = \alpha_{i} + G\xi + Z_{i}\zeta + \varepsilon_{i}$$
⁽¹¹⁾

where *endog*_i represents the j^{th} endogenous variable (M, y and z), **G** is a vector of all exogenous variables from Eq. (9), **Z** is a vector of instruments for the j^{th} endogenous variable. ε_i is the equation specific error term and ξ and ζ' are model parameters. Following Wooldridge (2010), we include residues (\overline{res}_j) from each of the reduced form equations as additional regressors in the main

outcome equation to both test and control for endogeneity.

We used the following IVs. For MT area (*M*), we used access to MT extension (dummy = 1) and distance to district center (*boma*). We would expect access to MT extension and distance to the district business center (where MT project offices and agro-dealers for inputs and MT implements are located) to affect farmer decisions to adopt MT, but not their cropland expansion decisions directly. For yield (*y*), we used whether the household head is polygamously married (dummy = 1), monogamously married (dummy = 1) and distance from homestead to main feeder road. For the shadow wage (*x*), we used whether the household head is polygamously or monogamously married (dummies = 1).

Suitable IVs should affect MT area, yield and shadow wages, but not expansion directly. We might argue that our IVs are only indirect measures of the factors that affect expansion such as labor availability that we controlled for in the estimation. We formally tested for IV relevance, i.e., that the IVs are significantly correlated to the endogenous variable(s).

Table B1 in appendix B shows that the IVs were relevant in each of their respective first stage equations (4.11 < F < 8.76). All the residues were statistically insignificant in the main outcome equations (Table B2). Thus, we failed to reject the null hypothesis that the (potentially endogenous) variables are exogenous.³ Therefore, in line with standard procedure (Ainembabazi and Angelsen 2016; Ricker-Gilbert, Jayne, and Chirwa 2011), our empirical estimation proceeded without considering prior endogeneity concerns. From now, M, y and z are considered exogenous in the estimation.

The second estimation issue is the censored nature of the outcome variable; only about 19% of the sampled farmers expanded cropland. Although Heckman and Tobit models are potential options, we used the double hurdle model which relaxes the Tobit assumptions by allowing the same or different factors to explain the decisions on (i) whether to expand or not, and (ii) the extent of expansion differently (Cragg 1971; Wooldridge 2010).

A third challenge concerns how to parcel out the effects of area under MT (our measure of MT effects) from those of crop yield on cropland expansion. Recall from our theoretical discussion that MT affect cropland expansion both through its effects on yield (higher profitability of expansion), but that the demands of more labor also increases the shadow wage, dampening or even reversing the outcome. In an attempt to isolate these two effects, we included MT area and crop yield in separate models. Yield in this case also includes the effects of MT on yield, but the size of area under MT better reflects labor demand related to MT.⁴

³ We arrived at similar conclusions using 2SLS and IV-Tobit models.

⁴ Our data do not allow us to explicitly isolate the contributions of MT to crop yield and to household labor supply and demand. We also tried to use 3SLS, but we dropped it because we could not prove the normality of the error covariance matrix.

3.3 Empirical specification

Following Wooldridge (2010), the first stage of the double hurdle model is a probit model of whether or not households expanded cropland, i.e., exp > 0:

$$P(\exp > 0 | \mathbf{G}) = \Phi(\mathbf{G}\gamma) = \alpha_1 + \mathbf{G}\psi + \varepsilon.$$
(12)

The second stage is a truncated normal regression of area expanded (A^*) conditional on exp>0 and G,

$$E(A^* | \exp > 0, \mathbf{G}) = \alpha_2 + \mathbf{G}\beta + \mu.$$
⁽¹³⁾

exp is a dummy = 1 if a household expanded cropland and A^* is the amount of new cropland in the 2013/2014 season. α_i are the intercept terms. *G* is a vector of all exogenous variables defined as before; β and γ are parameters to be estimated while ε and μ are error terms from the first stage and second stage respectively. The overall expected value of $E(A^*)$ is a product of the expected values for Eq. (12) and (13) and the first derivatives of these equations measure the marginal or average partial effects (APEs). Readers are referred to Wooldridge (2010) for a thorough discussion. We estimated the two steps of the double hurdle simultaneously with maximum likelihood, following Burke (2009).

4. DATA SOURCES

The data used in this study are from an extensive household survey conducted in 2014. Sampling proceeded at three levels. First, we selected Nyimba, Mumbwa and Mpika districts based on exposure to forest conservation interventions, agriculturally productive area and prevalence of shifting cultivation systems. Second, we randomly sampled 10 villages per district using the most recent village lists per district. Third, we used up-to-date village registers obtained from village leaders to select 12-15 households, randomly from each village for interview. This gave sub-samples of 120 from Mpika and Nyimba districts, 128 from Mumbwa, and 368 households in total. Figure 1 shows the location of survey areas.

Data were collected using structured questionnaires and face-to-face interviews. The survey collected detailed information on household demographics, agricultural (including tillage methods) and off-farm activities, yield, labor and other input use, asset holdings and sources of income. Specifically, the survey asked households whether they expanded their cropland in the 2013/2014 season. Those who expanded provided the size of the new (additional) area, reasons for expanding, the main crop(s) grown, where they expanded into, and who among household members made the decision. Similarly, those who did not expand provided reasons.

The main crops grown in the study areas include maize, cotton, groundnuts, sunflower, soybeans, mixed beans and cassava. Table 1 defines the main variables and presents summary statistics.



Figure 1. Spatial Location of Survey Areas in Zambia

Source: Authors

Variable description	Name	Mean	SD	Min	Max
Dependent variable					
Expanded cropland into $2013/2014$ season (yes =1)	$A > A^0$	0.19	0.40	0.00	1.00
Self-reported cropland expansion area (ha)	A^*	0.14	0.39	0.00	4.05
Potentially endogenous independent variables					
Land under MT (ha)	Μ	0.13	0.47	0.00	5.67
Shadow wage	Z	0.48	0.21	0.19	2.27
Aggregate crop yield (kg/ha) ⁺	у	1874.76	1229.54	190.00	6900.00
Exogenous Independent variables					
Distance to protected forest (km)	h	13.50	11.75	0.50	55.00
Accessed subsidy (yes $= 1$)	h	0.30	0.46	0.00	1.00
Adult equivalents	h	4.96	2.01	0.74	10.26
Farm size, next of expansion (ha)	A^0	4.29	6.93	0.11	100.00
Age, head of household	h	45.04	13.89	19.00	89.00
Male head of household (yes $= 1$)	h	0.77	0.42	0.00	1.00
Education, head of household (years)	h	6.37	3.23	0.00	18.00
Value of assets (ZMW'000)*	h	2.24	13.84	0.00	250.00
Some land on title (yes $= 1$)	h	0.04	0.20	0.00	1.00
Used hybrid seed and/fertilizer (yes = 1)	Х	0.78	0.41	0.00	1.00
Maize (output) price (ZMW/kg)	p_y	1.85	0.67	0.34	4.67
Fertilizer and seed (input) price (ZMW/kg)	p_x	4.50	1.41	1.15	13.12
A member earned off farm work (yes = 1)	l_0	0.09	0.28	0.00	1.00
Mumbwa district (yes=1)		0.37	0.48	0.00	1.00
Nyimba district (yes=1)		0.34	0.48	0.00	1.00
Mpika district (yes=1)		0.29	0.46	0.00	1.00
Additional instruments (IVs)					
Distance to district center (km)		29.04	24.47	1.00	95.00
Distance to feeder road (km)		2.56	7.55	0.00	90.00
Polygamy (yes $= 1$)		0.03	0.18	0.00	1.00
Monogamy (yes $= 1$)		0.71	0.45	0.00	1.00
Accessed MT extension (yes $= 1$)		0.59	0.49	0.00	1.00

Table 1. Variables Used in Regression Models and Summary Statistics

Notes: +For all crops, but mainly maize and used as a proxy for expected yield at the time farmers made the decision to expand in the 2013/2014 season. The sample reduced to 350 after dropping 18 households who had zero harvest because they only cultivated cassava during the survey period. (1 USD = 6.2 ZMK at the time of the survey.)

5. RESULTS

5.1 Where and Why Do Smallholders Expand Cropland?

Not all cropland expansion causes deforestation, and we asked respondents who expanded cropland about what land parcels were expanded, what land they expanded into, and - if fallows - how old the fallows were.

Overall, about 24% of the respondents expanded cropland during the survey reference period (i.e., between 2012/2013 and 2013/2014 agricultural seasons). Among these, 75% and 25% expanded into uncultivated land (forest) and fallows, respectively. Thus, when we define deforestation as cropland expansion into virgin forests or fallowed land older than 15 years, about 19% of the sampled households are involved in deforestation (Table 1).⁵ We base our empirical analysis on this second definition of expansion. Of all fallow land brought back into cultivation, only about a quarter had been in fallow for more than 15 years, a sign of shortening fallow periods in the study areas.

There is no significant difference in cropland expansion between MT adopters and non-adopters. Within each group, 18-19% of the households expanded their cropping area into forests during the last agricultural season. Among adopters, the average cropland cleared per household was 0.10 ha, while non-adopters cleared larger parcels on average - about 0.15 ha.

About 79% of land-expanding households did so for maize production. This finding corroborates those in Babigumira et al. (2014), who found that most of the cropland expansion in Africa is for maize production – the staple in many Sub-Saharan African countries.

Most households in the sample stated that they expanded to increase production to meet subsistence requirements (Figure 2). Other reasons included improved market access, declining land productivity, settling in new area (out-migration), improved access to inputs through government subsidies and clearing to secure title.



Figure 2. Reasons Farmers Expanded Cropland into the 2013/2014 Season

Source: Authors

⁵ We follow FAO (2015) and define forests as land parcels larger than 0.5 hectares and not in agricultural use, with tree canopy cover of more than 10% and that these trees should reach a minimum height of 5 meters in situ. This definition includes primary and secondary forest, native or exotic, as well as closed and open forest (e.g., woodlands).

Among households who did not expand cropland, the majority (68%) stated that they did not expand due to lack of resources (labor and cash) or means to do so. About 21% said there was no land to expand to, and 11% stated that they did not have any need to expand cropland.

5.2 Are Households That Expanded Cropland Different from Those That Did Not?

We checked for systematic differences between households that expanded cropland in the 2013/2014 season and those that did not. Table 2 shows, using t-tests, that households who expanded cropland had significantly larger farms (6.0 ha vs. 3.9 ha) before expansion, higher asset endowments and used improved inputs (a possible indicator of better market access). Among those that expanded, the share of male-headed households is higher (indicating the critical role of access to family (male) labor). At district level, Nyimba had the largest proportion of land-expanding households.

Although the differences are insignificant, results in Table 2 also show that the average crop yield was higher among households who expanded. A larger proportion of these households received input subsidies⁶, and they were on average more educated and had younger household heads.

	Expanded cropland into 2013/14 season			
	No	Yes	t-stat	
	Mean	Mean	(no-yes)	Sign. level
Land under MT	0.14	0.13	0.01	
Distance to protected forest	13.63	12.99	0.40	
Shadow wage	0.47	0.51	-1.63	
Yield	1,862.36	1,925.27	-0.38	
Adult equivalents	4.9	5.22	-1.2	
Farm size, next of expansion (ha)	3.88	5.99	-3.01	***
Age, head of household	45.48	43.22	1.22	
Male head of household	0.75	0.86	-1.91	*
Education, head of household	6.27	6.77	-1.15	
Value of assets (ZMW'000)	1.37	5.76	-2.38	**
Land on title	0.05	0.01	1.21	
Improved inputs	0.16	0.25	-1.76	*
Subsidy	0.29	0.36	-1.2	
Output price	1.84	1.93	-1.01	
Input price	4.51	4.46	0.29	
Off farm work	0.09	0.1	-0.32	
Mumbwa district (yes=1)	0.37	0.35	0.34	
Nyimba district (yes=1)	0.32	0.45	-2.09	**
Mpika district (yes=1)	0.31	0.20	1.81	*

Table 2. Bivariate Mean Comparisons of Key Variables between Households WhoExpanded Cropland into the 2013/2014 Season and Those That Did Not

Notes: ***, **, *: significant at 1%, 5% and 10%.

⁶ The effect of subsidies on expansion requires a separate study.



Figure 3. Quadratic Predictions of the Effects Minimum Tillage Adoption Intensity (A) Yield (B), Farm Size, (C) and Value of Household Assets (D) on Cropland Expansion

Source: Authors

We further explored the bivariate relationships between cropland expansion and key variables: area under MT, farm size, household wealth and yield. For each variable of interest, we estimated a simple bivariate quadratic regression to predict its effects on expansion.⁷ Figure 3 suggests a weak relationship between land cultivated under MT (adoption-intensity) and expansion. In line with *a priori* expectations, higher yield, farm size and household wealth are correlated positively and mostly significantly with expansion. However, bivariate analysis fails to control for correlations across explanatory variables and their possible endogeneity, which leads us to the econometric analysis of next section.

5.3 Econometric Results

We estimated several specifications of the double hurdle model. We report results from two models with the area under MT and yield variables included separately. These two models better capture and isolate the different channels through which MT may affect expansion, namely through yield and labor effects. Because the results are similar, we will focus mainly on model 1 and only refer briefly to model 2. The estimation was done with standard errors clustered at the village level. As we could not reject exogeneity, estimation proceeded without correcting for endogeneity. Table 3 presents the average partial effects (APEs) for the probability of expansion, extent of expansion conditional on expansion (i.e., among households who expanded), and the overall effects for the model with area under MT (model 1) in columns 1-3. The APEs for the model with yield (model 2) are given in columns 4-6.

⁷ We used the *qfitci* command in Stata.

Model 1 shows that the size of area under MT is negatively correlated to the probability and extent of cropland expansion, although these results are only significant among households who already expanded (column 2). However, the overall effect is insignificant (columns 3).

	Model 1			Model 2		
	(1)	(2)	(3)	(4)	(5)	(6)
	Participation	Participation	Overall	Participation	Participation	Overall
	APEs	APEs	APEs	APEs	APEs	APEs
MT area (ha)	-0.027	-0.494**	-0.113	-	-	-
	(0.057)	(0.201)	(0.081)	-	-	-
Yield (kg/ha) /100	-	-	-	2.20E-04	0.008*	0.002
	-	-	-	(0.002)	(0.004)	(0.002)
Shadow wage	0.154	1.347**	0.357**	0.139	0.188	0.122
	(0.112)	(0.575)	(0.164)	(0.112)	(0.387)	(0.165)
Adult equivalents	0.007	0.124***	0.028***	0.006	0.102***	0.023*
-	(0.010)	(0.028)	(0.011)	(0.010)	(0.036)	(0.013)
Land size, net of expansion	0.002	-0.022	-0.003	0.002	-0.023	-0.003
	(0.007)	(0.014)	(0.007)	(0.007)	(0.015)	(0.008)
Age household head	-0.003*	-0.011***	-0.004**	-0.003*	-0.012**	-0.004**
	(0.001)	(0.004)	(0.001)	(0.001)	(0.005)	(0.002)
Male household head (yes $= 1$)	0.059	0.016	0.040	0.058	-0.030	0.030
	(0.041)	(0.135)	(0.046)	(0.041)	(0.168)	(0.048)
Education household head	0.010	0.020	0.010*	0.010	0.028*	0.012*
	(0.007)	(0.014)	(0.006)	(0.007)	(0.014)	(0.007)
Asset value (ZMW) /1000	0.004	0.012	0.005	0.004	0.012	0.005
	(0.005)	(0.021)	(0.007)	(0.005)	(0.020)	(0.007)
Some land on title (yes=1)	-0.095	-1.046***	-0.262**	-0.097	-1.135***	-0.276*
	(0.104)	(0.310)	(0.120)	(0.105)	(0.390)	(0.155)
Improved inputs (yes $= 1$)	0.160**	-0.177	0.067	0.161**	-0.169	0.068
	(0.074)	(0.195)	(0.054)	(0.079)	(0.201)	(0.060)
Govt. input subsidy (yes $= 1$)	0.039	-0.030	0.019	0.037	0.018	0.026
	(0.053)	(0.117)	(0.042)	(0.053)	(0.115)	(0.044)
Output price per kg	-0.016	0.027	-0.005	-0.016	0.084	0.006
	(0.032)	(0.084)	(0.026)	(0.036)	(0.096)	(0.031)
Input price per kg	-0.036**	0.003	-0.022	-0.037**	-0.033	-0.029
	(0.017)	(0.061)	(0.017)	(0.017)	(0.071)	(0.018)
Off farm income (yes=1)	0.019	-0.020	0.008	0.016	-0.279	-0.043
	(0.070)	(0.146)	(0.064)	(0.072)	(0.192)	(0.077)
Dist. to protected forest (Km)	0.001	0.003	0.001	0.001	0.007	0.002
	(0.004)	(0.004)	(0.002)	(0.004)	(0.006)	(0.002)
Mumbwa district (yes=1)	0.057	0.503**	0.133**	0.053	0.884**	0.201**
	(0.091)	(0.220)	(0.063)	(0.091)	(0.363)	(0.082)
Nyimba district (yes=1)	0.209**	0.530**	0.234***	0.209**	0.837**	0.289***
	(0.097)	(0.234)	(0.062)	(0.098)	(0.339)	(0.079)
Sigma (o)	0.60***			0.73***		
Log pseudo likelihood	-173.91			-179.87		
Observations	350	69	350	350	69	350

Table 3. Double Hurdle Average Partial Effects of Factors Influencing Cropland Expansion.

Notes: ***, **, * significant at 1%, 5% and 10% respectively. Bootstrapped standard errors in parenthesis (300 replications). APEs refer to average partial effects.

This is in line with our descriptive results in Table 2 and Figure 3, where the size of land under MT did not seem to be correlated with cropland expansion overall, but it was weakly correlated among adopters. Adopters cleared less cropland than non-adopters did. Further, this result is qualitatively in agreement with our theoretical results, which suggest that the overall effects of MT on expansion are indeterminate *a priori*, as labor and yield effects pull in different directions.

Using improved inputs (inorganic fertilizers and hybrid seed) increases the likelihood of expansion by 16 percentage points (columns 1, Table 3), while an additional year in the age of the household head and a unit increase in input prices reduce the likelihood of expansion by 0.3 and 4.0 percentage points, respectively. These results are statistically significant at the 5% level.

Higher crop yield (model 2), shadow wage, education of the household head and adult equivalents are positively correlated with expansion among households who already expanded, and these results are significant at 1-10% levels (columns 2 and 5). For example, all else equal, doubling yield is predicted to increase expansion by about 80m². Secure land tenure reduces the extent of expansion (Models 1 and 2, columns 2 and 5).

The overall effects suggest that labor availability (adult equivalents) and shadow wage stimulate expansion while age of the household head and having some land on secure tenure could reduce expansion. There are significant location effects: farmers in the more densely populated Mumbwa and Nyimba districts (relative to those in Mpika district) were more likely to expand cropland.

6. DISCUSSION

6.1 Minimum Tillage and Deforestation

The main result of this paper is that higher MT adoption intensity does not significantly affect cropland expansion. An adoption intensity of MT averaging 0.13 ha or about 8% of cultivated land per farm household shows that MT is not the dominant tillage method at household level, and this could at least partly explain our results.

The overall results are in agreement with Byerlee, Stevenson, and Villoria (2014), who conclude that technology-driven intensification alone is unlikely to reduce cropland expansion without improved governance and incentives to preserve nature. They also echo earlier synthesis of case studies on the impacts of improved agricultural technologies on deforestation (Angelsen and Kaimowitz 2001).

However, MT adoption intensity is negatively correlated with expansion among households who already expanded. We can use our theoretical results and surmise that this captures the effects of MT on labor allocation. MT is labor intensive, at least in the short run, and labor scarcity may reduce expansion due to the higher demand for (family) labor among MT adopters. The net effects of MT on cropland expansion will depend on whether its yield or labor effects dominate - an issue that would require further investigation. (Note that the yield and labor effects are not equivalent to the substitution and income effects discussed in the theory section.) Because we are not able to investigate fully the labor intensity of MT in our data, this result should be interpreted with caution.

6.2 Other Drivers of Deforestation

Our finding that a higher opportunity cost of labor stimulates expansion runs counter to economic intuition and to our theoretical results. One explanation might, however, be found in the way the shadow wages are calculated, although the model is commonly used. The formula of Eq. (10) includes total agricultural production; a high shadow wage may therefore reflect that the household has certain characteristics (including unobservables) that increase productivity – also from recently cleared land. Additionally, this finding reflects that labor markets are missing and imperfect in the study areas. It also suggests that household labor is crucial for agricultural production – including land expansion – and that households employ more labor on-farm than supplying off-farm. This result also renders support to the inverse farm size – productivity relationships owing to the high labor/land ratio on small farms (Ainembabazi and Angelsen 2016).

The positive and significant effect of adult equivalents on expansion is in line with our *a priori* expectations about the role of access to family labor to facilitate expansion. This result is also in line with Babigumira et al. (2014) who found that households with more male labor were also more likely to expand cropland. The negative correlation between age of the household head and the likelihood, and extent of expansion is line with the life cycle theory and also corroborates findings in Babigumira et al. (2014). Households headed by younger household heads were more likely to expand, both due to higher physical strength and due to the need to invest in more land at early stages in the life of the household.

Results showing that higher yield stimulates cropland expansion (among households who already expanded) support long-standing arguments suggesting that higher crop productivity may stimulate deforestation (Angelsen and Kaimowitz 2001; Balmford, Green, and Scharlemann 2005; Rudel et al. 2009). Thus, technological-driven intensification on its own may stimulate expansion if market conditions are favorable (Byerlee, Stevenson, and Villoria (2014).

The negative effect of seed and fertilizer prices on the probability of cropland expansion highlight the importance of market factors as suggested in Babigumira et al. (2014). Our result that proximity to protected forests does not influence expansion decisions reinforces the need to improve local forest management institutions, which are generally weak in Zambia (Mulenga, Nkonde, and Ngoma 2015).

Tenure security may have contradictory effects on cropland expansion. It can make investments in existing land more secure, and help making intensification more attractive relative to expansion. If, however, farmers clear forests to claim tenure rights and titles, more secure tenure can spur forest clearing (Angelsen 1999). We find that secure land tenure reduces cropland expansion, suggesting that the first effect is stronger. This is in line with our findings that only 1% of respondents reported that they cleared forests to secure title. It is also in line with other studies suggesting that secure tenure facilitates farm level investments in land improvements in Ethiopia (Deininger and Jin 2006), Kenya (Kabubo-Mariara 2007), Zambia (Smith 2004) and Africa, in general (Place 2009) – which in turn would reduce the need for expansion – at least in the short run. Place and Otsuka (2000) found that conversion of forested land to agriculture was highest under customary (insecure) tenure in Uganda. However, because we measure tenure as a dummy, and we do not investigate further its potential endogeneity, caution is needed when interpreting the tenure results.

Finally, our findings suggesting that farmers in Mumbwa and Nyimba districts were more likely to expand relative to those in Mpika district highlight differing land pressures across the districts: the population density in Mpika is 5 persons/km² compared to 10 and 8 in Mumbwa and Nyimba, respectively. Only 19% of the respondents expanded cropland, highlighting that increasing land scarcity in Zambia is setting constraints to farmers' options (Chamberlin, Jayne, and Headey 2014).

7. CONCLUSION

This paper assessed the effects of minimum tillage and other proximate factors on cropland expansion. We developed a theoretical non-separable household model with limited off farm work opportunities and tested hypotheses using household survey data for the 2013/2014 agricultural season in Zambia. Future research should assess the effects of the full conservation agriculture package on deforestation, and develop long-term panel data sets to assess its effects on labor allocation and productivity, and cropland expansion. About one-fifth of smallholder households in our sample expanded cropland into forests clearing an average of 0.14 ha over one year. Its low adoption intensity, averaging 0.13 ha or 8% of cultivated land per farm household, however, suggests that it is not the dominant tillage method.

Minimum tillage is both yield augmenting and labor intensive, thus its net effects on expansion are ambiguous, *a priori*. In our dataset, we do not find evidence of lower expansion due to the adoption of minimum tillage practices overall, but among households who already expanded. The yield enhancing effect might, under the "right" conditions lead to more forest encroachment. Thus, we conclude in line with Byerlee, Stevenson, and Villoria (2014) that policies aimed to improve agricultural yields such as minimum tillage may exacerbate deforestation without concomitant forest conservation measures such as direct control of cropland expansion. Given the dual challenge facing African agriculture of both raising yield and farm incomes, while adapting to and mitigating climate change, conservation agriculture practices such as minimum tillage could be part of the overall policy package.

REFERENCES

- Ainembabazi, J. H. and Angelsen, A. (2016). Land Inheritance and Market Transactions in Uganda. *Land Economics*, 92 (1): 28-56.
- Alix-Garcia, J. M., Shapiro, E. N. and Sims, K. R. (2012). Forest conservation and slippage: Evidence from Mexico's national payments for ecosystem services program. *Land Economics*, 88 (4): 613-638.
- Angelsen, A. (1999). Agricultural expansion and deforestation: modelling the impact of population, market forces and property rights. *Journal of Development Economics*, 58 (1): 185-218.
- Angelsen, A. and Kaimowitz, D. (2001). Introduction: the role of agricultural technologies in Tropical deforestation. In Angelsen, A. and Kaimowitz, D. (eds) *Agricultural technologies and tropical deforestation*. Wallingford, Oxon, UK: CABI Publishing.
- Angelsen, A. (2010). Policies for reduced deforestation and their impact on agricultural production. *Proceedings of the National Academy of Sciences*, 107 (46): 19639-19644.
- Babigumira, R., Angelsen, A., Buis, M., Bauch, S., Sunderland, T. and Wunder, S. (2014). Forest Clearing in Rural Livelihoods: Household-Level Global-Comparative Evidence. World Development, 64: S67-S79.
- Balmford, A., Green, R. E. and Scharlemann, J. P. W. (2005). Sparing land for nature: exploring the potential impact of changes in agricultural yield on the area needed for crop production. *Global Change Biology*, 11 (10): 1594-1605.
- Barbier, E. B. and Burgess, J. C. (1997). The Economics of Tropical Forest Land Use Options. Land *Economics*, 73 (2): 174-195.
- Barbier, E. B. and Burgess, J. C. (2001). The Economics of Tropical Deforestation. *Journal of Economic Surveys*, 15 (3): 413-433.
- Burke, W. J. (2009). Fitting and interpreting Cragg's tobit alternative using Stata. *Stata Journal*, 9 (4): 584-592.
- Byerlee, D., Stevenson, J. and Villoria, N. (2014). Does intensification slow crop land expansion or encourage deforestation? *Global Food Security*, 3 (2): 92-98.
- Chamberlin, J., Jayne, T. S. and Headey, D. (2014). Scarcity amidst abundance? Reassessing the potential for cropland expansion in Africa. *Food Policy*, 48: 51-65.
- Cragg, J. G. (1971). Some statistical models for limited dependent variables with application to the demand for durable goods. *Econometrica: Journal of the Econometric Society:* 829-844.
- De Janvry, A. and Sadoulet, E. (2010). Agriculture for development in Sub-Saharan Africa: An update. *African Journal of Agricultural and Resource Economics*, 5 (1): 194-204.
- Deininger, K. and Jin, S. (2006). Tenure security and land-related investment: Evidence from Ethiopia. *European Economic Review*, 50 (5): 1245-1277.
- Ehui, S. K. and Hertel, T. W. (1989). Deforestation and agricultural productivity in the Côte d'Ivoire. *American Journal of Agricultural Economics*, 71 (3): 703-711.
- Ewers, R. M., Scharlemann, J. P. W., Balmford, A. and Green, R. E. (2009). Do increases in agricultural yield spare land for nature? *Global Change Biology*, 15 (7): 1716-1726.

- FAO. (2015). *Global forest resources assessment 2015*: Food and Agriculture Organization of the United Nations.
- Gibbs, H. K., Ruesch, A. S., Achard, F., Clayton, M. K., Holmgren, P., Ramankutty, N. and Foley, J. A. (2010). Tropical forests were the primary sources of new agricultural land in the 1980s and 1990s. *Proceedings of the National Academy of Sciences*, 107 (38): 16732-16737.
- Giller, K. E., Witter, E., Corbeels, M. and Tittonell, P. (2009). Conservation agriculture and smallholder farming in Africa: The heretics' view. *Field Crops Research* 114: 23–34.
- Govaerts, B., Verhulst, N., Castellanos-Navarrete, A., Sayre, K. D., Dixon, J. and Dendooven, L. (2009). Conservation agriculture and soil carbon sequestration: between myth and farmer reality. *Critical Reviews in Plant Sciences*, 28 (3): 97-122.
- Haggblade, S. and Tembo, G. (2003). Development, diffusion and impact of conservation farming in Zambia. *Food Security Research Project working paper # 8*. Lusaka: Food Security Research Project.
- Hallegatte, S., Bangalore, M., Bonzanigo, L., Fay, M., Kane, T., Narloch, U., Rozenberg, J., Treguer, D. and Vogt-Schilb, A. (2016). Shock waves: Managing the impacts of climate change on poverty *Climate change and development series*. Washington, DC: World Bank.
- Holden, S. T. (1997). Adjustment Policies, Peasant Household Resource Allocation and Deforestation in Northern Zambia: An Overview and Some Policy Conclusions. Forum for Development Studies, 24 (1): 117-134.
- IPCC. (2013). Climate change 2013: the physical science basis: Working Group I contribution to the Fifth assessment report of the Intergovernmental Panel on Climate Change. In Stocker, T. F. and D. Qin, G.-K. P., M. Tignor, S.K. Allen, J. Boschung, A. Nauels, Y. Xia, V. Bex and P.M. Midgley (eds). Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA.
- IPCC. (2014). Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part B: Regional Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change In Barros, V. R., C.B. Field, D.J. Dokken, M.D. Mastrandrea, K.J. Mach, T.E. Bilir, M. Chatterjee, K.L. Ebi, Y.O. Estrada, R.C. Genova, B. Girma, E.S. Kissel, A.N. Levy, S. MacCracken, P.R. Mastrandrea, and L.L. White (ed.). United Kingdom and New York, NY, USA. 688 pp.
- Jacoby, H. G. (1993). Shadow Wages and Peasant Family Labour Supply: An Econometric Application to the Peruvian Sierra. *The Review of Economic Studies*, 60 (4): 903-921.
- Kabubo-Mariara, J. (2007). Land conservation and tenure security in Kenya: Boserup's hypothesis revisited. *Ecological Economics*, 64 (1): 25-35.
- Kalinda, T., Bwalya, S., Munkosha, J. and Siampale, A. (2013). An appraisal of forest resources in zambia using the integrated land use assessment (ILUA) survey data. *Research Journal of Environmental and Earth Sciences*, 5: 619-630.
- Maertens, M., Zeller, M. and Birner, R. (2006). Sustainable agricultural intensification in forest frontier areas. *Agricultural Economics*, 34 (2): 197-206.

- Mulenga, B. P., Nkonde, C. and Ngoma, H. (2015). Does Customary Land Tenure System Encourage Local Forestry Management in Zambia? A Focus on Wood Fuel. *LAPRI working paper number 95*. Lusaka, Zambia: IAPRI.
- Pagiola, S. and Holden, S. (2001). Farm household intensification decisions and the environment. In Lee, D. R. and Barrett, C. B. (eds) *Tradeoffs or Synergies? Agricultural Intensification, Economic Development and the Environment.* New York: CABI Publishing.
- Phelps, J., Carrasco, L. R., Webb, E. L., Koh, L. P. and Pascual, U. (2013). Agricultural intensification escalates future conservation costs. *Proceedings of the National Academy of Sciences*, 110 (19): 7601-7606.
- Place, F. and Otsuka, K. (2000). Population Pressure, Land Tenure, and Tree Resource Management in Uganda. *Land Economics*, 76 (2): 233-251.
- Place, F. (2009). Land Tenure and Agricultural Productivity in Africa: A Comparative Analysis of the Economics Literature and Recent Policy Strategies and Reforms. World Development, 37 (8): 1326-1336.
- Powlson, D. S., Stirling, C. M., Jat, M. L., Gerard, B. G., Palm, C. A., Sanchez, P. A. and Cassman, K. G. (2014). Limited potential of no-till agriculture for climate change mitigation. *Nature Clim. Change*, 4 (8): 678-683.
- Powlson, D. S., Stirling, C. M., Thierfelder, C., White, R. P. and Jat, M. L. (2016). Does conservation agriculture deliver climate change mitigation through soil carbon sequestration in tropical agroecosystems? *Agriculture, Ecosystems and Environment*, 220: 164-174.
- Ricker-Gilbert, J., Jayne, T. S. and Chirwa, E. (2011). Subsidies and Crowding Out: A Double-Hurdle Model of Fertilizer Demand in Malawi. *American Journal of Agricultural Economics*, 93 (1): 26-42.
- Rudel, T. K., Schneider, L., Uriarte, M., Turner, B. L., DeFries, R., Lawrence, D., Geoghegan, J., Hecht, S., Ickowitz, A., Lambin, E. F., et al. (2009). Agricultural intensification and changes in cultivated areas, 1970–2005. *Proceedings of the National Academy of Sciences*, 106 (49): 20675-20680.
- Rudel, T. K. (2013). The national determinants of deforestation in Sub-Saharan Africa. *Philosophical Transactions of the Royal Society of London B: Biological Sciences*, 368 (1625).
- Singh, I., Squire, L. and Strauss, J. (1986). Agricultural household models: extensions, applications, and policy: Johns Hopkins University Press.
- Smith, R. E. (2004). Land Tenure, Fixed Investment, and Farm Productivity: Evidence from Zambia's Southern Province. *World Development*, 32 (10): 1641-1661.
- Tachibana, T., Nguyen, T. M. and Otsuka, K. (2001). Agricultural intensification versus extensification: a case study of deforestation in the northern-hill region of Vietnam. *Journal of Environmental Economics and Management*, 41 (1): 44-69.
- Thierfelder, C. and Wall, P. C. (2010). Investigating conservation agriculture (CA) systems in Zambia and Zimbabwe to mitigate future effects of climate change. *Journal of Crop Improvement*, 24 (2): 113-121.
- Thierfelder, C., Rusinamhodzi, L., Ngwira, A. R., Mupangwa, W., Nyagumbo, I., Kassie, G. T. and Cairns, J. E. (2015). Conservation agriculture in Southern Africa: Advances in knowledge. *Renewable Agriculture and Food Systems*, 30 (04): 328-348.

- UNEP. (2013). The emissions gap report 2013: United Nations Environmental Programme (UNEP), Nairobi.
- VandenBygaart, A. J. (2016). The myth that no-till can mitigate global climate change. Agriculture, Ecosystems & Environment, 216: 98-99.
- Vinya, R., Syampungani, S., Kasumu, E. C., Monde, C. and Kasubika, R. (2011). Preliminary Study on the Drivers of Deforestation and Potential for REDD+ in Zambia. *A consultancy report prepared for Forestry Department and FAO under the national UN-REDD+ Programme Ministry of Lands & Natural Resources.* Lusaka, Zambia.
- Wooldridge, J. M. (2010). *Econometric Analysis of Cross Section and Panel Data*. 2nd ed. MA, Cambidge: MIT Press.

APPENDICES

A. PROOF OF PROPOSITION

Appendix A provides proofs for the proposition using the first order conditions (FOCs) in Eq. (5) - (7). Because household behavior in our theoretical model is reflected through changes in the shadow wage (z) as given by Eq. (8), we first define the marginal effects of changes in leisure (l) and consumption (c) on z and then move on to show the full derivations for the first and second order derivations used in the comparative statics. The final step applies Cramer's rule.

Define the marginal effects of l and c on z as:

$$\begin{aligned} \frac{\partial z}{\partial l} &= z_l = \frac{\left(\frac{\partial^2 U_l}{\partial l^2} \times \frac{\partial U}{\partial c}\right) - \left(\frac{\partial U}{\partial l} \times \frac{\partial^2 U_c}{\partial l}\right)}{U_c^2} \\ &= \frac{U_{ll} U_c - U_{cl} U_l}{U_c^2} = \frac{U_{ll} U_c}{U_c^2} = \frac{U_{ll}}{U_c} < 0 \\ \frac{\partial z}{\partial c} &= z_c = \frac{\left(\frac{\partial^2 U_l}{\partial c} \times \frac{\partial U}{\partial c}\right) - \left(\frac{\partial U}{\partial l} \times \frac{\partial^2 U_c}{\partial c^2}\right)}{U_c^2} \\ &= \frac{U_{lc} U_c - U_{cc} U_l}{U_c^2} = -\frac{U_{cc} U_l}{U_c^2} > 0 \end{aligned}$$
(A1)

since U_{lc} , $U_{cl} = 0$; U_{ll} , $U_{cc} < 0$. Thus, z decreases with more leisure but increases with more consumption due to diminishing marginal utility, (cf. Eq. (8)).

The Langrage for the problem in Eq. (1) - (3) can be written as

$$L = U(c,l) + \mu \left[c - p_y f(l^a, M, A; X) + vM + d(A - A_o) + p_x X - wl^o - E \right]$$

+ $\lambda \left[T - l - t(A - A_o)^r - l^a - \overline{l^o} \right]$ (A2)

The first order conditions (FOCs) for the choice variables *l*, *c*, *M* and *A* are given by:

$$\partial L / \partial A = \mu \left(-p_y f_A + d \right) - \lambda \left(rt(A - A_o)^{r-1} \right) = 0 \tag{A3}$$

$$\partial L / \partial l^{a} = -\mu \left(p_{y} f_{l^{a}} \right) - \lambda = 0 \tag{A4}$$

$$\partial L / \partial M = \mu \left(-p_y f_M + v \right) = 0 \tag{A5}$$

$$\partial L / \partial c = U_c(c,l) + \mu = 0 \tag{A6}$$

$$\partial L / \partial l = U_l(c, l) - \lambda = 0. \tag{A7}$$

Using Eq. (A6) and (A7) to eliminate the shadow values of consumption (μ) and labor (λ), the FOCs for the choice variables, l', A and M in Eq. (5) – (7) can be written in their complete form as:⁸

$$L_{l^{a}}: p_{y}f_{l^{a}}(l^{a}, A, M) - \frac{U_{l}(c, l)}{U_{c}(c, l)} = 0$$

$$= p_{y}f_{l^{a}}(l^{a}, A, M) - \frac{U_{l}(p_{y}f(l^{a}, A, M) - vM - d(A - A_{0}) - p_{x}\mathbf{X} + E, l)}{U_{c}(p_{y}f(l^{a}, A, M) - vM - d(A - A_{0}) - p_{x}\mathbf{X} + E, l)} = 0$$
(A8)

$$L_{A}: p_{y}f_{A}(l^{a}, A, M) - d - \frac{U_{l}(c, l)}{U_{c}(c, l)}[rt(A - A_{o})^{r-1}] = 0$$
(A9)

$$L_{M} : p_{y} f_{M} (l^{a}, A, M) - v = 0$$
(A10)
where $l = T - t(A - A_{o})^{r-1} - l^{a} - \overline{l^{o}}$.

To simplify notation in the following derivations, we drop the *a* superscript in l^a and use *m* instead of *M* from henceforth. We implement the derivations using one FOC at a time for the variables of interest l, A, M, p_y, A_o and v. Similar results can be obtained by total differentiation.

Differentiating Eq. (A8) with respect to l, A, M, p_y, A_o, v and using results in (A1) to simplify yields:

$$\frac{\partial^{2} L_{l}}{\partial L_{l}^{2}} = p_{y} f_{ll} - \left[-\frac{U_{lc} p_{y} f_{l} U_{c} - U_{cl} U_{l}}{U_{c}^{2}} - \frac{U_{ll} U_{c} - U_{cc} p_{y} f_{l} U_{l}}{U_{c}^{2}} \right]
= p_{y} f_{ll} - \left[-\frac{\left(U_{lc} p_{y} f_{l} - U_{ll} \right) U_{c}}{U_{c}^{2}} - \frac{\left(U_{cc} p_{y} f_{l} - U_{cl} \right) U_{l}}{U_{c}^{2}} \right]
= p_{y} f_{ll} + \frac{U_{ll} U_{c}}{U_{c}^{2}} + \frac{U_{cc} U_{l}}{U_{c}^{2}} p_{y} f_{l}$$

$$= p_{y} f_{ll} + \frac{U_{ll}}{U_{c}} + \frac{U_{cc} U_{l}}{U_{c}^{2}} p_{y} f_{l}
= p_{y} f_{ll} - z_{l} - z_{c} p_{y} f_{l} = p_{y} f_{ll} - z_{l^{a}}$$
(A11)

where

⁸ Note that the FOCs can also be obtained by substituting Eq. (2) and (3) into Eq. (1).

$$\frac{\partial^{2} L_{l}}{\partial L_{l} \partial m} = p_{y} f_{lm} - \left[\frac{U_{lc} \left(p_{y} f_{m} - v \right) U_{c} - U_{cc} \left(p_{y} f_{m} - v \right) U_{l}}{U_{c}^{2}} \right]
= p_{y} f_{lm} - \left[\frac{\overline{U_{lc} p_{y} f_{m} U_{c}} - U_{cc} U_{l} \left(p_{y} f_{m} - v \right)}{U_{c}^{2}} \right]
= p_{y} f_{lm} + \frac{U_{cc} U_{l}}{U_{c}^{2}} \left(p_{y} f_{m} - v \right) = p_{y} f_{lm} + z_{c} \left(p_{y} f_{m} - v \right) = p_{y} f_{lm}$$
(A12)

$$\frac{\partial^{2} L_{l}}{\partial L_{l} \partial A} = p_{y} f_{lA} - \left[-\frac{U_{ll} tr(A - A_{o})^{r-1} U_{c}}{U_{c}^{2}} - \frac{U_{cc}(p_{y} f_{A} - d) U_{l}}{U_{c}^{2}} \right]$$

$$= p_{y} f_{lA} + \frac{U_{cc} U_{l}}{U_{c}^{2}} (p_{y} f_{A} - d) + \frac{U_{ll} U_{l}}{U_{c}^{2}} (tr(A - A_{o})^{r-1})$$

$$= p_{y} f_{lA} - z_{c} (p_{y} f_{A} - d) - z_{l} (tr(A - A_{o})^{r-1}) = p_{y} f_{lA} - z_{A}$$
(A13)

Since $(p_y f_m - v) = 0$ by Eq. (A10).

$$\frac{\partial^2 L_l}{\partial L_l \partial p_y} = f_l - \left[\frac{y U_{lc} U_c}{U_c^2} - \frac{U_{cc} U_l y}{U_c^2} \right] \\
= f_l + \frac{U_{cc} U_l}{U_c^2} y = f_l - z_c y,$$
(A14)

where y = f(l, A, M).

$$\frac{\partial^2 L_l}{\partial L_l \partial v} = -\left[\frac{U_{lc} m U_c - U_{cl} U_l}{U_c^2} - \frac{U_{cc} m U_l}{U_c^2}\right]$$
$$= -\left[\frac{\overline{U_{lc} m U_c} - U_{cl} U_l}{U_c^2} - \frac{U_{cc} m U_l}{U_c^2}\right]$$
$$= z_c m$$
(A15)

$$\frac{\partial^{2} L_{l}}{\partial L_{l} \partial A_{0}} = \left[-\frac{\left(U_{ll} \left(tr \left(A - A_{o} \right)^{r-1} \right) \right) U_{c}}{U_{c}^{2}} - \frac{U_{cc} dU_{l}}{U_{c}^{2}} \right] \\
= -\frac{U_{ll} U_{c}}{U_{c}^{2}} \left(tr \left(A - A_{o} \right)^{r-1} \right) \\
= -z_{l} \left(tr \left(A - A_{o} \right)^{r-1} \right) - z_{c} d$$
(A16)

Differentiating Eq. (A9) with respect to l, A, M, p_y, A_o, v and using results in (A1) to simplify yields:

$$\frac{\partial^2 L_A}{\partial L_A \partial l} = p_y f_{Al} + (z_l - z_c (p_y f_l))\beta$$
(A17)

where $\beta = tr(A - A_o)^{r-1} > 0$.

$$\frac{\partial^2 L_A}{\partial L_A^2} = p_y f_{AA} - z_A \alpha \tag{A18}$$

where Z_A is given in Eq. (A11) and $\alpha = (r-1)rt(A-A_o)^{r-2} > 0$.

$$\frac{\partial^2 L_A}{\partial L_A \partial A_o} = z_{A_o} \left[\left(r - 1 \right) r t \left(A - A_o \right)^{r-2} \right]$$
(A19)

where $z_{A_o} = z_l (rt) (A - A_o)^{r-1} - z_c d$

$$\frac{\partial^2 L_A}{\partial L_A \partial v} = z_c m \beta \tag{A20}$$

where, as before $\beta = tr(A - A_o)^{r-1} > 0$.

$$\frac{\partial^2 L_A}{\partial L_A \partial M} = p_y f_{AM} - z_c \left(p_y f_m - v \right) = p_y f_{AM}$$
(A21)

since $p_y f_m - v = 0$ by Eq. (7)

$$\frac{\partial^2 L_A}{\partial L_A \partial p_y} = f_A - z_c y \tag{A22}$$

where y = f(l, A, M).

Differentiating Eq. (A10) with respect to l, A, M, p_y, A_o, v and using results in (A1) to simplify yields:

$$\frac{\partial^2 L_m}{\partial L_m \partial l} = p_y f_{ml} \tag{A23}$$

$$\frac{\partial^2 L_m}{\partial L_m \partial A} = p_y f_{mA} \tag{A24}$$

$$\frac{\partial^2 L_m}{\partial L_m^2} = p_y f_{mm} \tag{A25}$$

$$\frac{\partial^2 L_m}{\partial L_m \partial A_o} = 0 \tag{A26}$$

$$\frac{\partial^2 L_m}{\partial L_m \partial v} = -1 \tag{A27}$$

$$\frac{\partial^2 L_m}{\partial L_m \partial p_y} = f_m \tag{A28}$$

Eq. (A11) - (A28) can be combined and arranged into matrix form as:

$$\begin{pmatrix} \frac{\partial^{2} L_{l}}{\partial L_{l}^{2}} & \frac{\partial^{2} L_{l}}{\partial L_{l} \partial A} & \frac{\partial^{2} L_{l}}{\partial L_{l} \partial m} \\ \frac{\partial^{2} L_{A}}{\partial L_{A} \partial l} & \frac{\partial^{2} L_{A}}{\partial L_{A}^{2}} & \frac{\partial^{2} L_{A}}{\partial L_{A} \partial m} \\ \frac{\partial^{2} L_{m}}{\partial L_{m} \partial l} & \frac{\partial^{2} L_{m}}{\partial L_{m} \partial A} & \frac{\partial^{2} L_{m}}{\partial L_{m}^{2}} \end{pmatrix} = - \begin{pmatrix} \frac{\partial^{2} L_{l}}{\partial L_{l} \partial p_{y}} & \frac{\partial^{2} L_{l}}{\partial L_{l} \partial A_{o}} & \frac{\partial^{2} L_{l}}{\partial L_{l} \partial v} \\ \frac{\partial^{2} L_{A}}{\partial L_{A} \partial p_{y}} & \frac{\partial^{2} L_{A}}{\partial L_{A} \partial A_{o}} & \frac{\partial^{2} L_{A}}{\partial L_{A} \partial v} \\ \frac{\partial^{2} L_{m}}{\partial L_{m} \partial P_{y}} & \frac{\partial^{2} L_{m}}{\partial L_{m} \partial A_{o}} & \frac{\partial^{2} L_{m}}{\partial L_{m} \partial v} \end{pmatrix}$$
(A29)

On the LHS is the coefficient matrix and the parameter matrix is on the RHS. Replacing the actual derivations into Eq. (A29) and rearranging yields:

$$\begin{pmatrix} p_{y}f_{ll} - z_{l^{a}} & p_{y}f_{lA} - z_{A} & p_{y}f_{lm} \\ p_{y}f_{Al} - z_{l^{a}}\beta & p_{y}f_{AA} - z_{A}\alpha & p_{y}f_{Am} \\ p_{y}f_{ml} & p_{y}f_{mA} & p_{y}f_{mm} \end{pmatrix} \begin{pmatrix} \partial l \\ \partial A \\ \partial m \end{pmatrix}$$

$$= \begin{pmatrix} -f_{l} + z_{c}y & z_{l}\left(tr(A - A_{o})^{r-1} + z_{c}d\right) & -z_{c}m \\ -f_{A} + z_{c}y & -z_{A_{0}}\left(r-1\right)tr(A - A_{o})^{r-2} & -z_{c}m\beta \\ -f_{m} & 0 & 1 \end{pmatrix} \begin{pmatrix} \partial p_{y} \\ \partial A_{o} \\ \partial v \end{pmatrix}$$
(A30)

Expanding the coefficient matrix on the LHS of Eq. (A30) using column 3, working through the resulting algebra and using earlier definitions for Z_A , β and α to simplify gives the following determinant:

$$|H| = p_{y}^{3} \begin{bmatrix} (f_{lA} - z_{l^{a}} p_{y}^{-1} \beta) (f_{Am} f_{ml} - f_{mm} (f_{lA} - z_{A} p_{y}^{-1})) \\ + (f_{AA} - Z_{A} \alpha p_{y}^{-1}) (-f_{Ml}^{2} + f_{mm} (f_{ll} - p_{y}^{-1} z_{l^{a}})) \\ - f_{MA}^{2} (f_{ll} - p_{y}^{-1} z_{l^{a}}) - f_{Am} f_{ml} (f_{lA} - z_{A} p_{y}^{-1}) \end{bmatrix} < 0$$
(A31)

Following earlier definitions in the production function and the differentiation results and using standard math rules for a maximum, it can be determined that the determinant in Eq. (A31) is negative.

Since the main interest is to determine the effects of the cost of implementing MT on cropland expansion, the outcome of interest is $\partial A / \partial v$, which can be evaluated using Cramer's rule:

$$\frac{\partial A}{\partial v} = \frac{|H_v|}{|H|} = \frac{\begin{pmatrix} -z_c m & p_y f_{lA} - z_A & p_y f_{lm} \\ -z_c m \beta & p_y f_{AA} - z_A \alpha & p_y f_{Am} \\ 1 & p_y f_{ml} & p_y f_{mm} \end{pmatrix}}{H}$$
(A32)

Stating with the substitution effect where z is held constant so that $z_l, z_c = 0$ (Angelsen 1999): $\frac{\partial A}{\partial v} = \frac{|H_v|}{|H|} < 0$; because $|H|_v > 0$ after dropping all terms with z_l and z_c in Eq. (A32), but |H| < 0 from Eq. (A31).

The income effect is always the opposite to the substitution effect (for normal goods) so that $\partial A / \partial v > 0$.⁹ Analytical results for the effects of other variables used in the empirical model can be obtained following similar procedures.

⁹ As discussed in section 2.3, the results for $\partial A / \partial M$ will have the opposite signs to those of $\partial A / \partial v$

B. ENDOGENEITY TESTS, ROBUSTNESS CHECKS AND PRODUCTION FUNCTION ESTIMATES

Appendix B reports the endogeneity and instrument relevance test results in Table B1. Table B2 reports the full double hurdle model results with the IV-control function and Table B3 report results for the Cobb-Douglas production function.

	(1)	(2)	(3)
	MT area	Yield	Shadow wage
Polygamy (yes $= 1$)	-	5.845*	0.231**
	-	(1.765)	(2.177)
Monogamy (yes $= 1$)	-	-	0.103***
	-	_	(3.600)
Distance feeder road	-	-0.139***	-
	-	(-3.038)	_
Distance district center	-0.003*	-0.074**	_
	(-1.890)	(-2 188)	_
Accessed MT extension (yes $= 1$)	0.088**	(2.100)	_
recessed wit extension (yes – 1)	(2, 230)		
A dult equivalents	0.011	0.129	0.008
Runt equivalents	(0.828)	(0.314)	(1243)
Land size not of expansion	(0.020)	0.237*	0.001
Land size, net of expansion	(1, 104)	(1.065)	(0.200)
A an household hand	(1.194)	(-1.903)	(0.290)
Age nousenoid nead	0.001	-0.029	-0.000
	(0.838)	(-0.689)	(-0.209)
Male household head (yes = 1)	0.050	-1.125	-0.086**
	(1.098)	(-0./83)	(-2.232)
Education household head	-0.004	0.516**	0.002
	(-0.377)	(2.533)	(0.637)
Asset value (ZMW) /1000	-0.009	0.217***	-0.001
	(-1.276)	(3.956)	(-0.410)
Some land on title (yes=1)	-0.038	5.578***	0.054*
	(-0.919)	(3.100)	(1.843)
Improved inputs (yes $= 1$)	0.064	0.045	0.034
	(1.237)	(0.028)	(1.681)
Govt. input subsidy (yes $= 1$)	0.011	-6.560***	-0.030
	(0.552)	(-6.505)	(-1.543)
Output price per kg	0.003	0.412	-0.013
	(0.289)	(0.670)	(-1.031)
Input price per kg	0.048	3.891	0.017
	(0.617)	(1.447)	(0.552)
Off farm income (yes=1)	0.002	-0.016	-0.001
	(0.527)	(-0.223)	(-1.150)
Dist. to protected forest (Km)	0.214***	0.584	0.152***
1	(3.399)	(0.265)	(4.611)
Mumbwa district (ves=1)	-0.016	-1.012	0.036
,	(-0.285)	(-0.502)	(1.167)
Constant	-0.173	25.760***	0.501***
	(-1.274)	(5.678)	(5.066)
	(()	(0.070)	(0.000)
Observations	350	350	350
R-squared	0.147	0.276	0137

Table B 1. First Stage - Reduced Form OLS Estimates for the Endogenous Variables

Notes: t-statistics in parenthesis; ***, **, * significant at 1%, 5% and 10% respectively; combined IV relevance test results – MT area F=4.12, p=0.03; Yield – F=5.26, p=0.01; shadow wage, F=8.75, p=0.00.

	Model 1 (MT area)	Model 2 (yield)		
	(1)	(2)	(4)	(5)
	Probit	Truncreg	Probit	Truncreg
MT area (ha)	-0.733	-0.078	-	-
	(1.643)	(1.692)	-	-
MT residues	0.631	-0.895	-	-
	(1.671)	(1.963)	-	_
Yield (kg/ha) /100	-	-	-0.082	0.064
	-	-	(0.067)	(0.126)
Yield residues	-	-	0.085	-0.044
	-	_	(0.067)	(0.124)
Shadow wage	1.232	1.028	2.320	-8.393
0	(2.960)	(5.407)	(2.988)	(6.568)
Shadow wage residues	-0.660	1.751	-1.865	9.555
0	(3.083)	(5.868)	(3.118)	(6.945)
Adult equivalents	0.038	0.230***	0.040	0.186**
Internet	(0.049)	(0.072)	(0.047)	(0.094)
Land size, net of expansion	0.021	-0.058	-0.013	-0.021
, i i i i i i i i i i i i i i i i i i i	(0.045)	(0.046)	(0.044)	(0.057)
Age household head	-0.008	-0.026**	-0.010*	-0.036**
0	(0.006)	(0.013)	(0.006)	(0.015)
Male household head (ves $= 1$)	0.268	-0.072	0.173	-0.311
, , , , , , , , , , , , , , , , , , ,	(0.167)	(0.232)	(0.197)	(0.453)
Education household head	0.040	0.042	0.077*	0.060
	(0.028)	(0.034)	(0.039)	(0.064)
Asset value (ZMW) /1000	0.009	0.030	0.033	0.013
	(0.027)	(0.038)	(0.023)	(0.054)
Some land on title (ves=1)	-0.343	-2.264***	-0.264	-3.038**
,	(0.434)	(0.835)	(0.384)	(1.367)
Improved inputs (ves $= 1$)	0.596	-0.268	1.028	-0.265
	(0.406)	(0.509)	(0.626)	(1.006)
Govt. input subsidy (ves $= 1$)	0.190	-0.047	0.146	0.357
1 , , , ,	(0.238)	(0.309)	(0.199)	(0.351)
Output price per kg	-0.031	-0.004	-0.515	0.201
	(0.160)	(0.300)	(0.493)	(0.859)
Input price per kg	-0.134*	-0.015	-0.087	-0.202
	(0.079)	(0.168)	(0.071)	(0.218)
Off farm income (yes=1)	0.092	-0.042	0.347	-0.668
	(0.294)	(0.324)	(0.382)	(0.814)
Dist. to protected forest (Km)	0.005	0.004	Ò.000	0.008
1	(0.015)	(0.010)	(0.016)	(0.020)
Mumbwa district (yes=1)	0.288	1.079	-0.032	3.307**
Ç ,	(0.803)	(1.311)	(0.611)	(1.644)
Nyimba district (yes=1)	0.832*	1.120*	Ò.706 ´	2.200**
	(0.479)	(0.618)	(0.456)	(0.957)
Constant	-2.129	-1.117	-0.662	1.687
	(1.720)	(3.233)	(2.537)	(5.032)
Sigma (σ)	0.584***	× /	0.689***	~ /
Observations	350	350	350	350

 Table B 2. Coefficient Estimates from Double Hurdle Models with Control Function-IV

 Approaches

Notes: Dependent variables- stage 1, expanded (yes/no), stage 2- area expanded (ha); robust standard errors in parenthesis; ***, **, * significant at 1%, 5% and 10% respectively; the non-significance of residues suggest that exogeneity of MT area, yield and shadow wage in the cropland expansion equation cannot be rejected. We arrived at similar conclusion even in a model which includes both MT area and yield (for brevity, we do not report these results here).

Estimation of Shadow Wages

We estimated household specific shadow wages following Jacoby (1993) and used a Cobb Douglas production function. The dependent variable is aggregate value of crop production for all crops grown by households in the sample. The main crops included maize, cotton, sunflower, groundnuts, dry beans, sorghum, millet and cassava. The value of production was computed using crop prices obtained during the survey. The explanatory variables were aggregated across crops per household, all transformed into logs. We also included as additional regressors, average household age, education and size to control for household specific factors that may influence agricultural production.

Homogeneity was imposed by normalizing the explanatory variables using the quantity of seed. Alternative estimations with trans-log did not yield better results.

The household specific shadow wage (z) was estimated using Eq. (10).

Coefficient	SE	t-statistic
0.432***	0.139	3.105
0.385***	0.053	7.221
0.303***	0.054	5.595
1.029	0.985	1.045
-0.001	0.004	-0.165
0.071***	0.020	3.548
4.6E-04	0.019	0.024
yes	yes	yes
2.307***	0.340	6.780
350		
0.659		
	Coefficient 0.432*** 0.385*** 0.303*** 1.029 -0.001 0.071*** 4.6E-04 yes 2.307*** 350 0.659	Coefficient SE 0.432*** 0.139 0.385*** 0.053 0.303*** 0.054 1.029 0.985 -0.001 0.004 0.071*** 0.020 4.6E-04 0.019 yes yes 2.307*** 0.340

Table B 3. Cobb Douglas Production Function Estimates

Notes: ***, **, * significant at 1%, 5% and 10% respectively, (Dependent variable, value of production).

www.feedthefuture.gov